Author’s response to reviews

Title: Quality analysis of smart phone sleep apps in China: Can apps be used to conveniently screen for obstructive sleep apnea at home?

Authors:

Zhaofeng Xu (474953549@qq.com)
Xin Luo (luoxingz@126.com)
Jianbo Shi (tsjbent@163.com)
Yin-yan Lai (Laiyy3@mail.sysu.edu.cn)

Version: 4 Date: 06 Sep 2019

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the comments concerning our manuscript entitled “Quality analysis of smart phone sleep apps in China: Can apps be used to conveniently screen for obstructive sleep apnea at home?” (ID: MIDM-D-19-00187). Those comments are all very valuable and very helpful for revising and improving our paper as well as the important guiding significance to our researchers. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to editors’ comments are listed below.

Response to Editor’s Comments:

1. Response to “Table 2. We note a reference to Table 2 within your manuscript. Please ensure that all figures, tables and supplementary/additional files are attached. Alternatively if you do not wish to include Table 2 please remove this reference”

Answer: We have placed Table 2 at the end of our manuscript. Table 2 was cited and should appear at line 291 in the manuscript.

2. Response to “Clean manuscript. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also
be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.”

Answer: We have removed all tracked changes and highlighting in our manuscript. This is the clean copy of our manuscript.

Once again, thank you very much for your comments and suggestions.

Your sincerely,

Yinyan Lai

2019.09.06